

A Service of

ZBU

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Maurizio, Roxana; Monsalvo, Ana Paula

# Working Paper Informality, labour transitions, and the livelihoods of workers in Latin America

WIDER Working Paper, No. 2021/19

**Provided in Cooperation with:** United Nations University (UNU), World Institute for Development Economics Research (WIDER)

*Suggested Citation:* Maurizio, Roxana; Monsalvo, Ana Paula (2021) : Informality, labour transitions, and the livelihoods of workers in Latin America, WIDER Working Paper, No. 2021/19, ISBN 978-92-9256-953-2, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2021/953-2

This Version is available at: https://hdl.handle.net/10419/229420

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



SA https://creativecommons.org/licenses/by-nc-sa/3.0/igo/







# WIDER Working Paper 2021/19

# Informality, labour transitions, and the livelihoods of workers in Latin America

Roxana Maurizio<sup>1</sup> and Ana Paula Monsalvo<sup>2</sup>

January 2021

United Nations University World Institute for Development Economics Research

wider.unu.edu

**Abstract:** This paper studies the incidence and heterogeneity of labour informality in six Latin American countries—Argentina, Brazil, Ecuador, Mexico, Paraguay, and Peru. We divide workers into five work statuses: formal wage-employed, formal self-employed, upper-tier informal wage-employed, lower-tier informal wage-employed, and informal self-employed. We evaluate the patterns of the occupational turnover between these work statuses and assess their impact on wage dynamics. In all the countries, wages are highest for formal workers and lowest for lower-tier informal jobs. The proportion of formal workers who maintain their work status of origin or move up the job ladder is significantly higher than those who transition into lower-paying work statuses. However, despite the high labour turnover experienced by lower-tier informal wage employees, most failed to move up the wage ladder. Education plays an important role, as it increases the probability of transitioning into a better job and, within informality, the chance of better wages.

Key words: informality, occupational turnover, education, wage dynamics, Latin America

JEL classification: J31, J46, N36

<sup>1</sup>Instituto Interdisciplinario de Economía Política (Universidad de Buenos Aires-CONICET), Buenos Aires, Argentina, roxanadmaurizio@gmail.com, <sup>2</sup>Universidad Nacional de General Sarmiento, Los Polvorines, Argentina, apmons@gmail.com

This study has been prepared within the UNU-WIDER project Transforming informal work and livelihoods.

Copyright © UNU-WIDER 2021 / Licensed under CC BY-NC-SA 3.0 IGO

Third-party content is not covered by the terms of the Creative Commons license. For permission to reuse third-party content, please verify terms at the source and, if applicable, contact the rights holder.

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9256-953-2

https://doi.org/10.35188/UNU-WIDER/2021/953-2

Typescript prepared by Lesley Ellen.

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland, Sweden, and the United Kingdom as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

# 1 Introduction

Since the 2000s, an increasing trend in labour formality has been observed in several Latin American countries. However, despite this positive evolution, informal employment continues to be one of the most distinctive characteristics in this region.

Informal employment is a complex and heterogeneous phenomenon. It encompasses wage earners and the self-employed, including employers and own-account workers. Likewise, informality can be found in both big firms and microenterprises. However, the existence of a broad group of informal workers is mainly associated with a high presence of small, unstructured firms that operate with very low levels of productivity and competitiveness.

At the same time, labour informality is not reduced by a single determinant, but rather by a combination of various factors. In particular, sustained job-generating growth enhances labour market predictability, thus favouring an increase in long-term, formal contracts. However, while sustained economic growth seems to be a necessary condition for the process of employment formalization, it is the interaction between that process and specific policies, such as labour inspection or incentives for formalization, that determines the results in terms of labour registration.

The main aim of this paper is to carry out an in-depth study of the structural characteristics of the different segments of informal and formal workers, and their changes, since the beginning of the new millennium in six Latin American countries—Argentina, Brazil, Ecuador, Mexico, Paraguay, and Peru—from a comparative and dynamic perspective. In particular, this study: (1) assesses the intensity of occupational turnover, focusing on the movements between informal and formal employment; (2) estimates the likelihood of different segments of informal workers moving to a better position inside informal employment or to a formal occupation; and (3) assesses the impacts of labour mobility on wage dynamics.

This selection of countries allows us to have a broad picture of the Latin American labour markets as they have occupational structures and dynamics that greatly differ from one another. Additionally, they account for about 70 per cent of the total population in the region. The data come from regular household surveys carried out by the national statistical institutes of the selected countries. Although these surveys are not longitudinal, their rotating panel sample allows flow data to be drawn from them.

Three aspects of this study are worth emphasizing. First, it employs information on labour transitions in addition to the more traditional cross-section analysis. In this regard, the paper contributes to the scarce but growing literature on occupational mobility in Latin America.

Second, instead of only focusing on the dichotomy between formality and informality, we examine the heterogeneity of informal work, distinguishing between lower-tier and upper-tier informal workers. This differentiation allows us to assess whether informality is a last resort for the first segment of workers to earn income, and whether it is preferred over formal employment by the second segment.

Third, a comparative analysis is carried out with the six Latin American countries. Their differences regarding labour market structures and overall level of development provide insights for the analysis.

The results show that in all the countries considered wages are the highest for formal workers and the lowest for lower-tier informal workers. Two contrasting labour mobility patterns are found: on the one hand, the proportion of formal workers who maintain their work status of origin or move up the job ladder is significantly higher than the proportion who transition into lower-paying work statuses; on the other hand, despite the high labour turnover experienced by lower-tier informal wage employees, most of them failed to move up the wage ladder.

Education strongly correlates both with a higher probability of transitioning to a better job and with higher wages, even in an informal occupation. In other words, informal workers who remain in this work status can still improve their wages through higher education.

These results are particularly important considering the high informality that still persists in the region and the close association between informality, low productivity, and low wages. Therefore, the findings provide useful guidance for the design of public policies aimed at reducing informality and enhancing the livelihoods of workers and families in the region.

The paper follows with the literature review about the characteristics of formality in Latin America. Section 3 details the sources of information and the identification of formal, upper-tier and lower-tier informal workers in the countries under study. Section 4 describes the methodologies used. Section 5 analyses the evolution of labour formality during the 2000s and the current composition of employment. Section 6 identifies the patterns and characteristics of employment transitions. Section 7 assesses their impacts on wage dynamics. Section 8 concludes.

# 2 Literature review

There are different arguments about the existence and persistence of informality in the developing world and, in particular, in Latin America. From the 'voluntary choice' perspective, workers and firms choose their optimal level of engagement with the institutions of state depending on their valuation of the net benefits associated with formality and the state's enforcement efforts.

In this line, De Soto (1986) argues that the origin and persistence of informality in developing countries is a response to the complicated and expensive regulations that production units, even small ones, must comply with. Similarly, Perry et al. (2007) conclude that workers and firms make cost–benefit analyses about whether to cross the line into formality, and frequently decide against it.

From another perspective, informality does not stem from workers' preference based on economic rationality but rather, in many cases, it is the only opportunity for employment (ECLAC 2008). At a more aggregate level, a highly heterogeneous production structure maintains the informal sector due to the limited capacity of higher-productivity sectors to fully absorb the labour force (Infante 2011).

However, the heterogeneity that exists within informality allows both approaches to be complementary rather than mutually exclusive. In particular, Fields (1990) reconciles these two views by identifying different segments within informality: 'easy-entry' and 'upper-tier'. While informality is a last resort for the first segment of workers, who seek to earn some income to survive, for the second segment, it is a preference over formal employment. Entry to the latter is restricted by financial capital and human capital requirements. Therefore, Fields (1990) emphasizes the duality that characterizes informal employment by differentiating between free-entry and restricted-entry components. For those workers excluded from formal employment and uppertier informal jobs, accepting a lower-paying informal position can be the only employment possibility.

The notion that informal employment is heterogeneous is also discussed by Kucera and Roncolato (2008). According to them: 'The question is not whether there exists some voluntary informal employment in developing countries, but rather how widespread it is and how this might vary for countries at different levels of development and for different workers, particularly men and women' (Kucera and Roncolato 2008: 2). Along this line, Kanbur (2017) discusses what exactly informality is, proposing an approach that moves away from considering informality as a uniform category. Other authors also recognize the existence of heterogeneity within informal employment (Basu et al. 2018; Chen 2012; De Vreyer and Roubaud 2013; Grimm et al. 2012).

Another widely studied dimension of informality has been the wage gap between formal and informal workers and the presence of wage segmentation. Tornarolli et al. (2014) confirm the existence of these gaps in almost all Latin American countries, both among women and men. Arias and Khamis (2008) also find significant wage penalties due to informality in Argentina. Tannuri-Pianto and Pianto (2002) use quantile wage regressions and selection models to analyse wages in formal, informal, and self-employed workers in Bolivia. Their results seem to confirm the existence of segmentation at the lower quantiles of the earnings distribution. However, findings at higher quantiles are more consistent with a voluntary choice by highly productive workers. Using different parametric and non-parametric econometric methods, Maurizio (2016) points to the existence of significant wage returns to formality in the six countries in the region—Argentina, Brazil, Chile, Mexico, Paraguay, and Uruguay. Moreover, the wage gaps are not constant across the income distribution but are larger at the lower extreme.

There are few studies on the characteristics of the labour formalization process observed in many Latin American countries over the 2000s. According to the International Labour Organization (ILO 2018), one of the most important transformations of Latin American labour markets is the process of formalization that has been observed since the beginning of the new millennium. Out of the 51 million jobs created in the region in the 2005–15 decade, 39 million were formal jobs, thus evidencing the reduction of the informal employment rate in this period.

Maurizio and Vázquez (2019) show that the process of labour formalization in some Latin American countries took place in a period of strong total employment growth, which resulted in the creation of a significant volume of new formal wage-earning occupations. For example, the number of formal jobs rose by almost 60 per cent in Argentina between 2003 and 2017, while total employment increased by 20 per cent. In Brazil, these numbers are 40 per cent and 20 per cent, respectively. In Ecuador, Paraguay, and Peru the number of registered jobs has more than doubled since the new millennium.

Bertranou et al. (2013) analyse the employment formalization in Argentina from 2003 to 2012 and the public policies associated with that process. They conclude that labour policies, such as labour inspection, must be accompanied by production, tax, social, and labour policies, along with stable economic growth in order to foster formality.

Along this line, Maurizio (2015) concludes that the interaction between job-generating growth and specific policies aimed at labour formalization determined the results observed in terms of labour registration in Argentina and Brazil. In addition, she finds that those informal workers that were initially located in the upper part of the distribution have the highest probability of becoming formal. This situation seems to be consistent with the fact that the process of formalization was more intense among those individuals that presented a 'better' vector of observable attributes. However, they ended up in the lower deciles of the formal wage earners' distribution. In other

words, they transitioned from the upper part of the income distribution of informal wage earners to the lower part of the income distribution of formal wage earners. Maurizio and Vázquez (2019) confirm these findings in six Latin American countries—Argentina, Brazil, Ecuador, Paraguay, and Peru.

Finally, some studies focus on the effects of the reduction of labour informality on labour income inequality (Amarante and Arim 2015; Beccaria et al. 2015; Beccaria et al. 2020; ECLAC and ILO 2014; Maurizio 2015; Maurizio and Vázquez 2015). One common finding is the positive correlation between the reduction in informality and the fall in wage inequality. ECLAC and ILO (2014) also study the impact of the formalization process on gender wage gaps finding a heterogeneous effect across countries. In Brazil, Ecuador, Panama, and Paraguay, the increase in formality narrowed the wage difference between men and women, given that among the latter the intensity of formalization was stronger. The formalization process was, however, unequalizing in the Plurinational State of Bolivia and Colombia.

# 3 Data and measurement of informality

# 3.1 Data

The data used in this paper come from regular household surveys carried out by the national statistical institutes of each country. Although these surveys are not longitudinal, their rotating panel sample allows flow data to be drawn from them. In such schemes, the total sample is divided into a certain number of household groups and each group remains in the sample for a given number of observation periods. Therefore, for each wave of the survey, one of these groups enters the sample while another one leaves. Consequently, it is possible to compare a given proportion of the sample between two or more waves.

For Argentina, the data source is the *Encuesta Permanente de Hogares* (EPH) carried out by the *Instituto Nacional de Estadística y Censos* (INDEC 2019). Micro-data are available for 31 urban areas. From 2003, after a major methodological change, the survey provides quarterly data. Households are interviewed in two successive quarters, stay out of the sample for the two following quarters, and are interviewed again for two more quarters. Therefore, the transitions that can be analysed are those that occur between two yearly observations (in the same quarter of two successive years) or between two successive quarters. The theoretical overlapping sample between one quarter and the same quarter of the following year, as well as between two successive quarters, is 50 per cent.

Brazil's data come from the *Pesquisa Mensal de Emprego* (PME) (IBGE 2016) and the *Pesquisa Nacional por Amostra de Domicilios Contínua* (PNADC) (IBGE 2020), conducted by the *Instituto Brasileiro de Geografia e Estadistica* (IBGE). The PME covers six major urban areas and provides monthly information. In this survey, 25 per cent of the sample is replaced every month. Households are observed during four consecutive months, stay out of the sample for eight months, and are interviewed again for another four months, allowing the construction of monthly and yearly panel data. Until 2003, each of these groups represented 25 per cent of the sample for eight woo of which are interviewed each month. This scheme implies a theoretical overlapping of 75 per cent of the sample in two successive years and of 50 per cent every 12 months, thus avoiding 'blind periods' (periods with no sample overlapping). The PME ended in 2015. The PNADC began to be carried out in 2012. It covers urban and rural areas. The survey is planned to have quarterly information collecting data of the whole sample during the three months. The sample rotation scheme adopted is 1–2 (5). In this scheme, the household is interviewed for one month and leaves

the sample for two consecutive months; this sequence is repeated five times for each household before leaving the sample.

The Encuesta Nacional de Empleo, Desempleo y Subempleo (ENEMDU) in Ecuador is carried out by the Instituto Nacional de Estadística y Censos (INEC 2018). It has a rotating sample scheme that allows quarterly and yearly panels to be built by interviewing each household up to four times. Data are representative of urban and rural areas. It collects information in March, June, September, and December every year in 23 provinces of the country. Households are interviewed in two consecutive quarters, stay out of the sample for two quarters, and are interviewed again for two additional quarters. The sample is divided into four groups which represent 25 per cent of the sample, ensuring an overlapping of 50 per cent during the same month of two consecutive years. Contrary to the procedures followed by other surveys, the ENEMDU replaces households that do not respond to the survey and thus it has a very low non-response rate (of around 3 per cent of the initial sample).

The *Encuesta Nacional de Ocupación y Empleo* (ENOE) is the source of data for Mexico. It is conducted by the *Instituto Nacional de Estadística y Geografía* (INEGI 2007) and has national representativeness. The sample rotation scheme involves keeping 80 per cent of the sample overlapped between quarters and rotating the remaining 20 per cent. Each household remains in the sample for five consecutive quarters before leaving the sample.

The Paraguayan *Encuesta Continua de Empleo* (ECE) is carried out by the *Dirección General de Estadística, Encuestas y Censos* (2017) and has national coverage. The scheme used corresponds to a panel sample, which implies visiting the same households a maximum of five times.

Finally, in the case of Peru, the *Encuesta Nacional de Hogares* (ENAHO), the regular household survey conducted by the *Instituto Nacional de Estadistica e Informática* (INEI 2020), also covers urban and rural areas. The ENAHO panel for the period under analysis includes two types of rotation scheme. The first one corresponds to the period 2002 to 2006 when the panel was fixed. The second type began in 2007; since then the rotation scheme renews the sample by 20 per cent each year.

The period under analysis corresponds to the period since the start of the new millennium. However, the specific years considered vary in each country according to data availability. In Argentina all years between 2003 and 2019 are analysed, 2003–19 for Brazil (2003–15 PME, 2015–19 PNADC), 2003–19 for Ecuador, 2002–19 for Peru, 2005–19 for Mexico, and 2010–17 for Paraguay.

To obtain datasets that could be compared between countries, we included one transition for each individual, based on a one-year interval between observations. This enables us, for instance, to assess whether each person remained employed, became unemployed, or left the labour force. In addition, yearly transitions between different work statuses are also identified.

Our study is restricted to male workers between 15 and 65 years of age and female workers between 15 and 60. The upper limits correspond to the compulsory retirement ages in the countries being analysed, and we have used them in an attempt to minimize the bias that might come from the exits of older individuals from the labour force. Those individuals for whom information was incomplete or inconsistent regarding personal or occupational variables were removed from the sample.

Since not all the surveys used in this study are representative of each country as a whole, and given that labour markets in rural areas and urban centres may behave differently, our analysis only

covers urban areas. To obtain sufficient observations, yearly panels have been pooled in each country, so the results are the average for the period.

One potential problem arises from the fact that not every labour transition can be captured when matching two observations with a one-year observation window. Given that a transition is identified by comparing these two waves, two or more symmetrical changes in the work status (or in other variables) might take place between those two moments without being captured by the observed data.

Another limitation of panel information is that the proportion of households and individuals that are actually interviewed at two successive moments can be smaller than the proportion of those that should be re-interviewed according to the sample rotation scheme. This loss of cases ('attrition') is caused by different factors, e.g. people abandoning the panel or difficulties in the data collection process. If this loss is not random, it can introduce bias in the sample. However, there was no available information in the databases that allowed us to differentiate loss of data due to attrition associated with the survey rotation scheme. Therefore, an attrition bias correction was not possible. An indirect way to assess potential attrition effects (although it is not enough proof to discard its impact) is to compare key descriptive statistics from panel and cross-section data. We computed the formality rate among different groups of workers and found no relevant differences between them (Table A1).

# 3.2 Identification of formal, upper-tier, and lower-tier informal workers

Labour informality is one of the categories of analysis that has greatly contributed to the characterization of labour conditions in Latin America.

The concept of the informal sector (IS) emerged for the first time in the early seventies, in the ILO's documents for African countries (ILO 1972). It was then developed in the Latin American region by the Regional Employment Program for Latin American and the Caribbean (PREALC as its acronym in Spanish), with the objective of explaining the growth of wide sectors of the population that were not able to participate in the processes of productive modernization through a formal labour market, in a context of relatively low levels of unemployment and light countercyclical behaviour.

Under this 'productive approach', informality reflects the inability of the economies of the region to generate enough formal jobs in relation to the growth of the labour force. Given the shortage or lack of social protection mechanisms that provide an income to those who cannot access a job, some individuals decide to embark on any activity that would allow them to obtain an income that is sometimes barely enough to survive on.

In the nineties, the ILO (1993) defined a 'productive informal sector unit' as one characterized by fixed assets that do not belong to the company but rather to their owners; therefore, it is often not possible to distinguish which expense should be borne by the company and which corresponds to the household. Also, labour relationships in these units are mainly based on personal and social ties. Given these characteristics, the IS is usually associated with small productive units with no clear separation between capital and labour and with low levels of productivity. For this approach, the functioning logic of enterprises in informal employment (IE) is survival more than accumulation. The jobs generated in this sector constitute the employment in the informal sector (EIS).

The 15th and 17th International Conference of Labour Statistics (ICLS) of ILO established the classification criteria for formal and informal workers: according to the 'productive approach', the

EIS is defined as the workers employed in small productive units which are not legally registered as firms, employ a reduced amount of capital, and make limited use of technology.

However, given that household surveys do not inquire in depth into the characteristics of the firms, the ILO suggests adopting a measurement criterion based on the combination of occupational categories, occupation groups defined according to job qualifications, and the size of the firm. In this way, it is possible to identify the two major components of the IS: (1) family units comprising own-account workers and family workers; and (2) microenterprises comprising employers and wage earners in establishments with less than five employees. In the case of independent workers, only those with no professional skills (approximated by those individuals with incomplete university studies) are considered as part of the IS, as an operational way to leave only independent workers with low productivity in this sector. The public sector is excluded from the IS.

Along with these conceptual developments based on a 'productive approach', IE is another concept that has developed in more recent years. Based on a 'legal approach', IE refers to a different dimension of informality because it focuses directly on job conditions. In particular, this approach associates informality with the evasion of labour regulations, defining IE as that of workers not covered by labour legislation.<sup>1</sup>

In recent years, the ILO has gone further in the distinction between IE and EIS. According to Hussmanns (2004: 2):

'Employment in the informal sector' and 'informal employment' are both measures that are useful for analytical and policy-making purposes, as they refer to different aspects of the 'informalisation' of employment and to different targets for policymaking. One of the two concepts cannot replace the other. However, the two concepts need to be defined and measured in such a way that they are consistent and that one can be clearly distinguished from the other. (Hussmanns 2004: 2)

The ILO recommendations indicate that wage earners 'are considered to have informal jobs if their employment relationship is, in law or in practice, not subject to national labour legislation, income taxation, social protection or entitlement to certain employment benefits'.<sup>2</sup> The empirical identification of informal wage jobs in each country is based on available information derived from household surveys.

By combining these two approaches (and based on comparable available data across countries), this study identifies five different work statuses among workers, as described in Box 1.

<sup>&</sup>lt;sup>1</sup> ILO (2002), Hussmanns (2004).

<sup>&</sup>lt;sup>2</sup> Hussmanns (2004: 6).

Box 1: Identification and measurement of five work statuses

Work status	Measurement
Formal wage employees	<ul> <li>Argentina: those who answer that their employers make payroll deductions to pay social security contributions.</li> <li>Brazil: those who have signed a labour contract.</li> <li>Paraguay, Mexico, and Peru: those enrolled in a pension system.</li> <li>Ecuador: those indicating that they receive social insurance from the job.</li> </ul>
Upper-tier informal wage employees	Informal wage employees working in firms with more than five employees
Lower-tier informal wage employees	Informal wage employees working in firms with up to five employees
Formal self- employed	Owners in enterprises with more than five employees, and professional own-account workers
Informal self- employed	Owners in enterprises with up to five employees, non- professional own-account workers and unpaid family workers.

Source: authors' elaboration.

When putting the legal approach into practice, we seek to make comparable the formal wageearners identification criterion, which does not necessarily imply the same empirical implementation in each country given that household surveys capture this dimension in different ways. However, comparability is not affected much because, although every country chooses different variables, they all share the same notion of informality, i.e. the non-fulfilment, or evasion, of the labour legislation and social security regulations.

# 4 Methodology

In addition to the descriptive analysis based on cross-sectional data, this study relies on econometric exercises from annual panel data. Starting with the dynamic analysis, the year-on-year transition matrices between different states (employed, unemployed, and inactive) and different work statuses are computed. Then, multinomial logit regressions are estimated in order to evaluate how these transitions vary according to observable workers' attributes. Finally, wage dynamic equations are performed to estimate the impact of the mobility across work statuses on labour income changes.

#### 4.1 Multinomial logit regressions

As mentioned, multinomial logit regressions are estimated to identify the observed characteristics associated with changes in the initial work status, differentiating those attributes positively correlated with transitions into higher-paying occupations from those correlated with higher transitions to lower-paying occupations. Specifically, from these regressions we estimate the probability of being in an employment status k at time t = 1 ( $S_{i1}$ ) depending on the initial work status at time t = 0 ( $S_{i0}$ ) and on observed individual characteristics (X<sub>i0</sub>).

The statistical model can be formulated as follows:

$$\pi_{i1k} = P(S_{i1} = k] S_{i0}, X_{i0}) = \frac{e^{S_{i0}\theta_k + X_{i0}\beta_k}}{1 + \sum_{j=1}^{K} e^{S_{i0}\theta_j + X_{i0}\beta_j}}$$
(1)

Where  $\pi_{i1k}$  represents the transition probability from the base status  $S_{i0}$  to status  $S_{i1} = k$  for the individual *i*. Thus, the  $\theta_k$  and  $\beta_k$  are the regression parameters associated with outcome status *k*.

Then, equation (1) can be expressed using the logit link for a generalized linear model in order to estimate the parameters

$$log\left(\frac{\pi_{i1k}}{\pi_{i1(k+1)}}\right) = S_{i0}\theta_k + X_{i0}\beta_k \tag{2}$$

We estimate employment transition equations between three work statuses: (1) formal wage employees and formal self-employed; (2) upper-tier informal wage employees; and (3) lower-tier informal wage employees and informal self-employed. In this way, we obtain three multinomial logit equations for each country under study, one for each work status, and three outcomes in each one: stay in the same work status or transit to one of the other two possible statuses.

For identifiability, the base category chosen stays at the initial work status in all cases.

#### 4.2 Wage dynamics regressions

After evaluating the characteristics associated with different patterns of labour mobility, we estimate the correlation between work status changes and wage dynamics, controlling for observable worker attributes. For this purpose, we regress the change in the logarithm of labour earnings between time t = 0 and time t = 1 ( $\Delta y_1$ ) on the individual's initial log earnings ( $y_0$ ), the initial worker characteristics like education, age, gender, branch of activity ( $X_0$ ), and a set of dummies to model transitions from the initial job status to each one of the remaining statuses ( $D_{ij}$ ). Our dynamic income model also considers year and region fixed effects (equation 3).

$$\Delta y_{1i} = \beta_{0i} + \beta_{1i} y_{0i} + \beta_{2i} X_{0i} + \sum_{j} \vartheta_{ji} D_{ji} + \mu_i$$
(3)

The subscripts *i* and *j* represent lower-tier informal, upper-tier informal, and formal work status, and  $i \neq j$ . The base category *i* means the worker did not change status between periods. The coefficients  $\vartheta_{ji}$  can be interpreted as the premium or penalty when the worker moves from i-work status to j-work status relative to the worker who remains in the same work status *i*.

# 5 Evolution of labour informality and labour composition in Latin America during the 2000s

## 5.1 The labour formalization process

As mentioned before, Latin America has experienced a reducing trend in the informality rate among wage employees since the beginning of the new millennium. In the six countries under study, labour formalization led to an increase of around 14 percentage points (pp) in Argentina and 8 pp in Brazil in the formality rate among salaried workers in the period under study. This process has been even more intense in Ecuador, Paraguay, and Peru, where the share of formal workers grew by between 17 pp and 23 pp. Mexico, in contrast, only experienced a slight increase, about 2 pp, between the two ends of the period (Figure 1).

The labour formalization process in these countries during the 2000s has been associated, on the one hand, with greater dynamism in the generation of new jobs in a macroeconomic context characterized by a relatively high and stable growth rate, and, on the other hand, with the implementation of specific public policies aimed at reducing the costs of informality through various incentive mechanisms.

Indeed, the business cycle is a relevant factor to consider when analysing the drivers of the decline in labour informality. There are theoretical arguments on both the demand and supply sides of the labour market that account for the countercyclical nature of informality. As mentioned earlier, sustained job-generating growth enhances labour market predictability, thus favouring an increase in long-term, formal contracts. In addition, a period of sustained growth in labour demand may also lower the expected probability of layoffs and consequently the probability of employers having to face the costs of firing a formal worker. Hence, the incentives to maintain informal labour relations are reduced. In this context, employers can benefit from productivity increases as a result of the accumulation of specific human capital. There is evidence regarding the positive impact of the economic cycle on formalization (Boeri and Garibaldi 2007; Bosch and Esteban-Pretel 2009; Bosch and Maloney 2010; Corsueil and Foguel 2012).

Even when the evidence supports the procyclical behaviour of labour formality, economic growth seems a necessary but insufficient condition. In particular, some of the specific policies implemented in these countries, such as incentives for employment formalization and labour inspections, may also account for the process of formalization. Some studies have found positive impacts of the reduction of employers' contributions or the simplification of administrative procedures to register workers on formal labour demand (Castillo et al. 2012; Delgado et al. 2007; Fajnzylber et al. 2009; Monteiro and Assunção 2012), while others have not found significant impacts (Chacaltana 2001, 2008; Cruces et al. 2010). In addition, the few studies evaluating the impact of inspections seem to confirm that they have a positive effect on labour market formality (Almeida and Carneiro 2009; Berg 2010; de Andrade et al. 2013; Ronconi 2010; World Bank 2012).

However, after the significant increase in labour formality in the countries under analysis (except in Mexico), this process has slowed down, stopped, or reversed in recent years, hand in hand with the weakening of the macroeconomic performance. As shown in Figure 1, the turning point seems to have been around 2014/15. In Argentina, after some stagnation between 2010 and 2015, the formality rate fell by 2 pp between that year and 2019. A similar situation is observed in Ecuador, with a slowdown in the increase in formality between 2012 and 2015 and a subsequent reduction of 4 pp over the last four years.

These two contrasting phases are also observed in Brazil where there was an increasing trend in formality until 2015 and a fall between 2015 and 2019 (-3.5 pp). After the strong formalization process in Peru and Paraguay until 2014 and 2015, respectively, the proportion of formal workers remained relatively constant. Mexico experienced a different process from the rest of the countries considered; initially, between 2007 and 2010, the formality rate reduced by 3 pp, then remained at this level until 2012, and then experienced a slight increase of 2 pp mainly between 2012 and 2014. As mentioned earlier, as a net result of these contrasting dynamics the formality rate only increased by 2 pp throughout the period considered.



Figure 1: Evolution in the formality rate among urban salaried employment during the 2000s

Source: authors' elaboration based on household surveys.

## 5.2 Labour composition

Behind these common patterns, the composition of employment is highly heterogeneous among the countries considered in this study, which gives greater worth to the comparative analysis. As we can see in Figure 2, Brazil exhibits the highest level of wage formality in salaried employment (73 per cent), followed by Argentina (65 per cent), Ecuador, Peru, and Mexico (about 60 per cent) and, finally, by Paraguay (54 per cent).

The ranking of countries, however, is somewhat different when considering the share of formality in total employment due to the dissimilar incidence of self-employment. While in Brazil and Argentina around half of the urban employed are formal wage earners, this drops to 44 per cent and 41 per cent in Mexico and Paraguay, respectively, and to 36 per cent in Peru and Ecuador. Therefore, these figures show that formal wage earners, those covered by labour regulation, represent a small portion (at most half) of the total urban employed. In Ecuador and Peru, this group of workers is even lower numerically than non-salaried workers. In all these cases, then, the more recent behaviour of labour informality is particularly worrisome given that this phenomenon is far from negligible in the region.

In order to have a more detailed look at the composition of employment, Table A2 presents the distribution of different groups of individuals across the five work statuses (formal wage employment, upper-tier informal wage employment, lower-tier informal wage employment, formal self-employment, and informal self-employment) unemployment, and labour inactivity.

As already mentioned, employment in Brazil is more formal than in the other countries considered. Together, formal wage employees and the formal self-employed account for about 35 per cent of the working-age population (58 per cent of total workers). At the other end, these two groups of workers represent only 21 per cent in Peru (30 per cent of total workers).

Except in Peru, formal wage employees are the biggest group among the five work statuses. In any case, the outlook is very alarming since this group of workers represents at most only half of total urban employment. This work status is followed by the informal self-employed. They represent around 20 per cent of the total workers in Argentina, about 25 per cent in Brazil, Mexico, and Paraguay, 31 per cent in Ecuador, and 40 per cent in Peru. Formal self-employed workers are a very small group, comprising 5 per cent or less of total employment in all countries.

The relevance of informal non-salaried workers in the Latin American labour market is more evident within total informal workers (even more so among self-employed workers), where they account for about 40 per cent in Argentina and Paraguay, about 50 per cent in Mexico, and more than 50 per cent in Brazil, Ecuador, and Peru. In turn, although in all countries they represent 80 per cent or more of the total self-employed workers, the proportion of formal self-employed workers is around 20 per cent in Argentina and Brazil, while it is 10 per cent or less in Ecuador and Paraguay.



#### Figure 2: Employment composition in six Latin American countries

- Formal wage earners
- Informal wage earners
- Non-wage earners

#### Source: authors' elaboration based on household surveys.

The number of lower-tier informal wage employed is larger than the number of upper-tier informal salaried workers. The gap between these two groups is particularly significant in Argentina, Ecuador, and Mexico, where the first group represents between 50 per cent and 70 per cent more than the second group.

In summary, with the exception of Peru, the ranking of work statuses according to their relative importance in total employment is as follows: formal wage employees, informal self-employed, lower-tier informal wage employees, upper-tier informal wage employees, and formal self-employed. In Peru the first two positions are exchanged, with the rest being the same (Figure 3).



Figure 3: Detailed employment composition in six Latin American countries

Note: countries are ranked by the proportion of formal wage employees in total employment.

Source: authors' elaboration based on household surveys.

#### 5.3 Individual characteristics of workers and wages according to their status

Table A3 shows the main personal attributes of workers according to their status. As found in the empirical literature for Latin American countries (Alaniz et al. 2020; Amarante and Arim 2015; Bertranou and Casanova 2013; Bertranou et al. 2013; ECLAC 2006, 2013; ILO 2014; Maurizio and Vázquez 2019), education level is highly and positively correlated with formality. In particular, formal workers are over-represented among workers with complete tertiary/university education (TC). However, informal non-salaried and lower-tier informal workers exhibit, on average, the lowest education levels (secondary incomplete (SI) or secondary complete/university incomplete (SC/TI)). Upper-tier informal wage employees are in between.

Except in Ecuador, women are over-represented among lower-tier informal wage employment. However, the distribution of the other work categories significantly differs between countries. In Argentina and Brazil, the higher female presence in the informal lower tier is offset by the lower proportion of women among informal self-employed workers. As a consequence, the composition of the remaining work categories by gender is relatively similar to the proportion of women and men in total employment. However, in Ecuador, Mexico, Paraguay, and Peru, women are overrepresented among the informal self-employed. In all countries, the proportion of men in uppertier informal positions is higher than the proportion in total employment. In addition, men are over-represented among formal wage employees only in Peru.

As previous studies (Bertranou and Maurizio 2011; Maurizio 2020) have shown, self-employment is not an 'entry door' to the labour market for young workers in Latin America. Rather, young people work mostly in informal salaried positions. However, the likelihood of being a formal employee or in informal self-employment grows linearly with age: older workers are overrepresented in these two work statuses. In addition, there is U-shaped behaviour in the association between age and formal salaried employment, where its incidence is highest among workers aged 25–45 years.

In order to have a broader description of employment in the six Latin American countries, Table A4 presents weekly hours and hourly wages (in local currencies) by work status. As the abovementioned empirical literature for the region has widely found, informal workers earn lower wages than formal workers. This is verified both among wage earners and among the self-employed.

In addition, among wage employees the mean hours worked are highest for formal workers in all countries, followed by upper-tier informal and then by lower-tier informal salaried workers. Therefore, the gaps in monthly labour income are even wider than those found in hourly labour income. Among the non-salaried workers, the differences in working hours between formal and informal are (except in Paraguay) less significant.

Figure 4 shows the monthly wages by work category. In all countries, formal self-employed workers are located at the upper tail of the wage distribution, followed by formal wage employees. Depending on the country, the third position is occupied by the upper-tier informal wage employees or by informal self-employed workers.





Source: authors' elaboration based on household surveys.

In order to test for statistically significant differences in average hourly earnings across these work statuses, Figure 5 presents the coefficient of the fitted one-way ANOVA model, where the dependent variable is the log hourly earnings of each group of workers and the covariates are the work status dummies. The base category is formal wage employment.

The results prove that the formal self-employed exhibit the highest average hourly earnings, followed by formal wage employees. This wage difference is significant in all countries. Except in Paraguay, informality is associated with a statistically significant earnings penalty, with lower-tier informal jobs being the lowest paid. The gap between upper-tier informal wage employees and the informal self-employed is significant in Argentina, Brazil, and Mexico.



Figure 5: ANOVA of mean hourly income across work status groups

Note: coefficient estimates and 95% confidence interval of fitted one-way ANOVA model of mean hourly labour income using formal wage employees as the base category.

Source: authors' elaboration based on household surveys.

Finally, Kernel density functions give us a more detailed picture not only of the position of each work status on the wage ladder, but also of within-group dispersion (Figure 6). It is clear that variance is greater within self-employed workers than within wage earners. Therefore, when a comparison is made between formal workers, the abovementioned higher average earnings of the formal self-employed compared to the formal salaried workers is due to the skewness towards the right of the distribution of the former compared with the latter.

A similar situation appears when comparing the informal self-employed with the worst-paying position, i.e. the lower-tier informal wage employees: even if, on average, the former have higher earnings, the lowest labour incomes are registered in this group.

From this non-conditional analysis, however, we cannot conclude that the wage differences indicate a true wage penalty for certain groups of workers as they may reflect the different composition of characteristics and personal attributes by work status. In addition, these results suggest that it is relevant to investigate the extent to which the different degree of within-group inequality is explained, at least in part, by a different return to individual attributes in each of these work statuses. The conditional analysis carried out in Section 7, based on econometric estimates, will provide insight on this issue.



Figure 6: Density functions of hourly wage by work status

Source: authors' elaboration based on household surveys.

To sum up, the distribution of workers in each work status is not random in any country. Primeaged workers and those with a higher educational level are more likely to be formal than the other workers. In turn, wages are statistically higher among formal than informal positions. Formal wage earners are, additionally, covered by labour legislation and receive social security benefits, such as paid vacation, maternity leave, and future contributory pension. Overall, this evidence suggests that formal work is the most preferred.

The following analysis will contribute to a better understanding of the characteristics and patterns of mobility of each work status.

# 6 Employment transitions

## 6.1 Transition matrices

Starting with the dynamic analysis, the year-on-year transition matrices between different states (employed, unemployed, and inactive) and different work statuses are analysed. Table A5 presents the average results for the whole period.

The first finding is that, as expected, formal workers in all the countries studied are more likely to remain in a formal job than the other workers. In particular, between 80 per cent and 90 per cent of initially formal wage earners were still formal workers one year later. The higher occupational stability among formal wage earners may be explained by the existence of firing costs as well as by the fact that they are more concentrated in big, more stable companies.

For the remaining 10–20 per cent, the most frequent destination after leaving a formal salaried position is upper-tier informal wage employment in Mexico, Paraguay, and Peru, or inactivity in Argentina, Brazil, and Ecuador. However, even in these last three countries, transitions between formality and upper-tier informality are common.

However, in all countries, very few formal employees go into lower-tier informal salaried employment and even fewer into formal self-employment (2 per cent at most). As formal selfemployment is made up of professional own-accounts and employees in firms with five or more employees, these types of transitions imply that formal employees becomes independent professionals or employers in medium or large firms. Therefore, it is not be easy for salaried employees without a university degree, in particular, to become employers in medium or large companies. This may explain why transitions to an informal self-employment position are more frequent, albeit very low. In any case, outflows from formal salaried occupations to informal wage positions are more important in all countries than those going into independent occupations.

Informal wage employees (considered to be both segments within this group) are the most mobile group of workers. Self-employed workers are in between. The lower labour stability of informal wage earners could be explained by the fact that they have low or no legal firing costs, thus making them attractive for employment in industries with unstable demand and for unstable occupations. Moreover, informal employees have a greater presence in small-scale firms which are regularly exposed to risks that make them more vulnerable. As they operate with low capital/labour ratios, it is easier to decide to interrupt economic activity.

Beyond this general outlook, in all countries, upper-tier informal wage employees are even more mobile than formal workers and lower-tier informal workers. Except in Paraguay, they are the most unstable group of workers. Only about one-third of them stay in this work status from year to year in Brazil, Ecuador, and Mexico, 40 per cent in Argentina and Peru, and 50 per cent in Paraguay. These figures compare with 44 per cent and up to 53 per cent of lower-tier informal workers. It is striking that a better work status, at least in terms of average earnings, is less stable than those considered to be of worse status.

However, upper-tier informal workers are more likely to transition into formal wage employment than into lower-tier informal salaried employment. In fact, in all countries, formal salaried employment is the most frequent destination for those workers. At least in part, this may be associated with the characteristics of the salaried formalization process observed in the region for most of the period under study. In particular, as found by Maurizio and Vázquez (2019) and Beccaria et al. (2020), a significant proportion of the new formal workers in Latin America actually became formal employees in the same job (in situ formalization); i.e. a worker becomes formal maintaining the same occupation between t and t + 1. In addition, a positive relationship is found between the probability of becoming formal and the size of the firm. Therefore, the low job stability of upper-tier informal workers did not always imply that they actually left their position, but that some of them became formalized in the initial job. This may explain part of the 'apparent' transitions from this work status to formality as this process was observed with greater intensity in medium and big companies.

Mobility patterns are very different in the case of lower-tier informal wage employees who, as shown before, are located on the lowest rung of the job ladder. While between 44 per cent and 53 per cent of them remain in this work status for at least one year, when leaving this position, they are more likely to go into informal self-employment than any other work status. Fewer than 10 per cent of these workers move to formal salaried employment.

Between 60 per cent and 70 per cent of those workers in an informal self-employed position (nonprofessional own-account, unpaid family worker, or employer in small enterprises) remain in this status over the following year. The most frequent destination for those workers who leave their initial occupation is inactivity. This was to be expected as the intermittence associated with certain typical occupations of such workers could imply that many do not actively seek a new job while 'waiting' for new demand for their services or products.

In contrast, most informal self-employed workers do not find formal jobs. Indeed, the percentage of transitions to formal jobs is extremely low (less than 5 per cent) in all countries. Those workers and inactive individuals are the two groups with the least probability of moving towards formality, even lower than the unemployed. Instead, their options are limited to precarious wage-earning positions or another form of own-account employment, probably also of low quality.

Therefore, transitions between the two categories of informality—informal self-employment and lower-tier salaried employment—are very frequent. In fact, for each of these categories, the other category is the main occupational destination after leaving the initial position.

Lastly, given that in Latin America, including the countries concerned, unemployment spells are relatively short—basically because of a lack or low coverage of unemployment insurance—only about 15 per cent to 27 per cent of individuals who were initially unemployed remained so one year later. In most cases, this reflects the high number of transitions to inactivity (20–30 per cent) and informal jobs (13–35 per cent). In all cases, it is evident that although the unemployed quickly leave this initial state, this does not translate into intense entries into formality, since in almost all cases (except in Mexico) less than 16 per cent of those unemployed in one year are in formal employment one year later. The same is the case for the inactive group, for whom the probability of entering the workforce through a formal occupation is very low. As mentioned earlier, transitions between informal, self-employment, and inactivity are very frequent in all countries.

This evidence allows us to evaluate whether informality is a persistent state or whether it is a steppingstone towards a better-paid occupation. The results suggest that there is significant labour turnover of workers from lower-tier informal salaried occupations. In fact, together with upper-tier informal wage employees, these are the more mobile categories of workers. In contrast, the informal self-employed exhibit a relatively high level (60–70 per cent) of job stability from year to year.

Complementary to this, the destinations after leaving the initial position also help to answer that question. Figure 7 summarizes the share of stayers for each work status, the share of transitions into a higher-paying job, and the share of transitions into a lower-paying job. This classification is

based on the mean hourly earnings of each status presented earlier. Therefore, unemployment and economic inactivity were not considered in these clusters.





Lower-tier informal employees

Source: authors' elaboration based on household surveys.

As the lower-tier informal wage employees are in the worst-paying work status, they only have two alternatives: remaining in this type of occupation or transitioning up the job ladder. More than a half (between 52 per cent and 67 per cent) are part of the first group, meaning that a lower share of informal workers improve their employment position by moving to a higher-paying occupation: about 30–35 per cent in Brazil and Paraguay, 40 per cent in Argentina and Mexico, and about 48 per cent in Ecuador and Peru. This suggests that an important group of workers who were in this position in year t were in a better work status in year t+1. However, even in the period under analysis, characterized by improvements in the labour markets and by a strong formalization process, most failed to move up the wage ladder. Furthermore, as shown before, only a small proportion (less than one-third) of workers transitioned from a lower-tier informal wage job to a formal salaried position.

The greater stability of informal self-employment positions (mostly non-professional own-account workers) together with the high level of transitions between these occupations and lower-tier informal wage employment result in a very low percentage of transitions to a higher-paying job status (less than 13 per cent).

As mentioned earlier, in all countries, upper-tier informal wage employees are more likely to transition into formal wage employment than into lower-tier informal salaried employment or informal self-employed. However, when we consider those who stay and those who transition into a lower work status, together they account for about two-thirds or more of total transitions.

Finally, for the two formal groups of workers (formal self-employed and formal wage employees) the percentage of those who maintain their initial work status (and those formal wage employees moving up the job ladder) is significantly higher than the share of workers transitioning into lower-paying work statuses (except in Paraguay in the case of formal self-employed).

# 6.2 Characteristics associated with work status transitions

This heterogenous overview of labour turnover suggests that, in addition to the work status of origin, observed personal attributes are also associated with different probabilities of transitioning into higher- or lower-paying occupations.

Table A6 presents the results of the multinomial logit regressions. As mentioned previously, for this analysis we collapsed the employment transition equations into three work statuses: (1) formal wage employees and formal self-employed; (2) upper-tier informal wage employees; and (3) lower-tier informal wage employees and informal self-employed.

With a few exceptions, education is highly correlated with movements from low-paid informal positions to better-paid jobs. In particular, higher education promotes transitions from lower-tier and upper-tier informal positions to formal positions. These results seem to be consistent with previous findings on the characteristics of labour formalization in Latin America. For example, Maurizio (2016) found that skilled workers benefited from this process more than other categories of workers.

Education also plays a role in improving the position within informality, as it tends to be correlated positively with transitions from lower-tier to upper-tier informal jobs. Conversely, having secondary or university education reduces the chances of leaving formality and moving towards either of the other two work statuses. Therefore, the positive correlation between education and formality is observed through two channels: it contributes to higher stability in formal jobs and it fosters entries to this type of job.

In addition, in almost all cases, the linear coefficient of age (used as a proxy of general labour experience) is statistically significant and negative, and the quadratic coefficient is positive. This means that the primed-aged are more stable in formal jobs than the other workers. The roles of education and age may be explained by the accumulation of specific human capital, which usually complements general human capital. Formal workers with higher levels of education tend to receive more specific training, and thus employers try to retain them.

Finally, gender is another dimension associated with different patterns of labour mobility. Except in Paraguay, men have a greater probability of moving to a better position than women. This seems consistent with the greater difficulties faced by women in the Latin American labour market, a stylized fact largely documented in the empirical literature for the region (Alaniz et al. 2020; Beccaria and Paz 2016; ECLAC and ILO 2019; ILO 2014). However, unexpectedly, in Brazil, Ecuador, and Mexico, men also experience some more-intensive movement in the opposite direction.

## 7 Wage dynamics

Finally, this section aims to evaluate the wage dynamics associated with labour turnover. Table A7 presents the results of the wage dynamic equation described in Section 4. For each work status of origin, we estimate the proportional change in wages between year t and year t+1 associated with changes in the initial status, controlled by a set of personal and job characteristics. Therefore, unlike in the previous (non-conditional) descriptive analysis, by following the same worker over a year, we can evaluate whether the particular labour transition experienced implies an increase or a fall in wages, all other observable variables being constant.

As in the previous section, we have grouped the previous five work statuses into three groups: (1) formal wage employees and formal self-employed; (2) upper-tier informal wage employees; and (3) lower-tier informal wage employees and informal self-employed. Each column in Table A7 indicates the work status of origin. The transition dummies take the value zero if the worker remains in the same position, and the value one if moving into another work status. Also, Figure 8 displays the estimated marginal effect on changes in log earnings by initial and destination employment status.

In almost all the countries except Paraguay, the econometric results confirm the findings from the descriptive analysis. The sign and the magnitude of the transition effect indicate that formal work status is preferred to upper-tier informal work, and the latter is preferred to a lower-tier informal job. In particular, transition from formality to any other work status implies a loss of wage. This happens with greater intensity when workers move to a lower-tier informal position or informal self-employment. In contrast, transitions from lower-tier informal employment to formality always increase wages. However, transitions from lower-tier to upper-tier imply a true wage premium only in Argentina, Brazil, and Peru. Finally, upper-tier informal workers are the 'intermediate' group, as they obtain an increase in wages when they move into a formal position, but they experience a wage reduction when they transit to lower-tier informal work.

#### Figure 8: Labour income dynamics



# Note: each point shows the estimated marginal effect on changes in log earnings by initial and destination employment state. The grey lines show the 95% confidence intervals.

#### Source: authors' elaboration based on household surveys.

The rest of the coefficients represent the effect on the wage changes where the worker remains in the same job status over both periods. In almost all countries (except Paraguay) the gender dimension is statistically significant: women experience a wage penalty in all work statuses, suggesting wage discrimination against them. It is interesting, however, to note that the magnitude of this grows as we move down the job ladder: the wage penalty reaches the highest value among lower-tier informal workers and the lowest level among formal workers. This contributes to intensifying the observed higher incidence of poverty and of the working-poor phenomenon among women in the region (Amarante and Colacce 2019; Maurizio 2018).

As expected, education is one of the most relevant dimensions in wage determination and is positively correlated with wages in all work statuses. Again, the returns to education are lower among formal workers than among low-paying informal workers. Age is also correlated with higher wages (except in Paraguay and Peru, where some coefficients were not statistically significant). In Argentina, Brazil, Ecuador, and Mexico, this variable reaches the highest (lowest) positive influence in lower-tier informal and informal self-employed work statuses (formal jobs).

Therefore, wage gaps associated with different observable variables appear to be lower in formality than in informality. This seems to be consistent with the abovementioned lower within-variance in the first group than in the second. Labour institutions, such as minimum wage or collective bargaining, and labour regulations could account for the lower wage dispersion among formal wage earners (Amarante and Arim 2015; Keifman and Maurizio 2014). In particular, to the extent that these institutions are binding and provide an income floor, they can reduce inequality within the group of workers covered by them.

At the same time, these results show that informal workers who remain in their position of origin can improve their wages by increasing their level of education.

Finally, the initial wage has a strong and negative impact on the change in earnings. That is, the higher the initial wage, the lower its real growth during the period under analysis. It is interesting to note that this is verified in a statistically significant way in each work status, but with even greater intensity in the lowest categories of the job ladder. This pattern seems to be consistent with the reducing trend in inequality observed in Latin America and, in particular, in these countries over the 2000s (Cornia 2012; ECLAC 2017; Lustig et al. 2013; Maurizio and Vázquez 2016).

# 8 Final remarks

This paper analysed, from a comparative and dynamic perspective, the heterogeneity of formal and informal employment in six Latin American countries: Argentina, Brazil, Ecuador, Mexico, Paraguay, and Peru. This selection of countries gives us a broad picture of the Latin American labour markets, as they have occupational structures and dynamics that greatly differ from one another. Additionally, they account for about 70 per cent of the total population in the region.

We distinguished five particular work statuses: formal wage-employed, formal self-employed, upper-tier informal wage-employed, lower-tier informal wage-employed, and informal self-employed. We evaluated the intensity and patterns of the labour turnover between these work statuses and assessed its impact on wages.

In all the countries, formal workers earn the highest wages and lower-tier informal workers earn the lowest. Two contrasting labour mobility patterns were found: on the one hand, the proportion of formal workers who maintain their work status of origin or move up the job ladder is significantly higher than the proportion who transition into lower-paying work statuses; on the other hand, despite the high labour turnover experienced by lower-tier informal wage employees, most of them failed to move up the wage ladder.

Education plays an important role by increasing the probability of transitioning to a better job and by increasing wages, even in informal occupations. In other words, informal workers who remain in this work status can improve their wages by increasing their level of education.

Different types of public policies are needed to improve labour conditions in Latin American countries. As shown, after the significant increase in labour formality in the countries under analysis (except Mexico), this process has slowed down, stopped, or reversed in recent years. This

is particularly worrisome given that labour informality is far from negligible in the region. Therefore, progress in employment formalization and the strengthening of labour institutions' policies are essential to enable jobs to become an effective mechanism to overcome poverty and achieve social protection.

In turn, efforts are required to increase investment in human capital. This should be complemented by a coherent system of training as well as other active labour policies.

In addition, to progress towards greater global efficiency and better labour conditions, there is a need for a medium- and long-run development path which generates continuous demand for greater human capital and enables the offer of education to be reflected in higher wages and better working conditions. It is also essential that the institutional design of these programmes ensures that they last over time, so that such programmes become permanent state policies to achieve both short- and long-term objectives.

Finally, all these policies should be framed within an economic development strategy based on an integrated productive structure leading to high efficiency and systemic competitiveness. Productive convergence within a framework of high productivity standards is a necessary condition for sustained growth, employment promotion, and wage increases over time.

#### References

- Alaniz, E., T. Gindling, C. Mata, and D. Rojas (2020). 'Transforming Informal Work and Livelihoods in Costa Rica and Nicaragua'. WIDER Working Paper 2020/100. Helsinki: UNU-WIDER. https://doi.org/10.35188/UNU-WIDER/2020/857-3
- Almeida, R., and P. Carneiro (2009). 'Enforcement of Labor Regulation and Firm Performance'. *Journal of Comparative Economics*, 37(1): 28–46. https://doi.org/10.1016/j.jce.2008.09.004
- Amarante, V., and R. Arim (2015). *Designaldad e Informalidad*. Un análisis de cinco experiencias latinoamericanas. CEPAL-Ministerio de Asuntos Exteriores de Noruega. https://doi.org/10.18356/43426b8e-es
- Amarante, V., and M. Colacce (2019). 'Poverty and Gender in Latin America'. Paper presented at Eighth ECINEQ Conference. Paris, 3–5 July.
- Arias, O., and M. Khamis (2008). 'Comparative Advantage, Segmentation and Informal Earnings: A Marginal Treatment Effects Approach'. IZA Discussion Paper 3916. Bonn: IZA.
- Basu, A.K., N. Chau, G. Fields, and R. Kanbur (2018). Job Creation in a Multi-sector Labour Market Model for Developing Economies'. Oxford Economic Papers, 71(1): 119–44. https://doi.org/10.1093/oep/gpy029
- Beccaria, L., R. Maurizio, and A. Monsalvo (2020). 'Beyond the Returns to Education: Distribute Impact of Urban Salaried Labour Formalization in Latin America During the 2000s'. *Journal of Development and Change*, (forthcoming).
- Beccaria, L., R. Maurizio, and G. Vázquez (2015). 'Recent Decline in Wage Inequality and Formalization of the Labor Market in Argentina'. *International Review of Applied Economics*, 29(5): 677–700. https://doi.org/10.1080/02692171.2015.1054369
- Beccaria, L., and J. Paz (2016). 'La inserción precaria de mujeres y los jóvenes en el mercado laboral en américa latina y sus consecuencias para instituciones laborales'. Documento de Trabajo. Santiago: CEPAL.
- Berg, J. (2010). 'Laws or Luck? Understanding Rising Formality in Brazil in the 2000s'. ILO Working Paper 5. Brasilia: ILO. https://doi.org/10.1057/9780230307834\_5

- Bertranou, F., and L. Casanova. (2013). Informalidad laboral en Argentina: Segmentos críticos y políticas para la formalización. Buenos Aires: Organización Internacional del Trabajo.
- Bertranou, F., L. Casanova, M. Jiménez, and M. Jiménez (2013). 'Informalidad, calidad del empleo y segmentación laboral en Argentina', Documento de Trabajo 2. Buenos Aires: Oficina de País de la OIT para la Argentina.
- Bertranou, F., and R. Maurizio (2011). Trabajadores Independientes, Mercado Laboral e Informalidad en Argentina. Buenos Aires: OIT.
- Boeri, T., and P. Garibaldi (2007). 'Shadow Sorting'. In J. Frankel, and C. Pissarides (eds.), NBER International Seminar on Macroeconomics 2005. Cambridge: MIT Press.
- Bosch, M., and J. Esteban-Pretel (2009). 'Cyclical Informality and Unemployment'. CIRJE Discussion Paper 613. Tokyo: CIRJE.
- Bosch, M., and W. Maloney (2010). 'Comparative Analysis of Labor Market Dynamics Using Markov Processes: An Application to Informality'. *Labour Economics*, 17(4): 621–31. https://doi.org/10.1016/j.labeco.2010.01.005
- Castillo, V., S. Rojo Brizuela, and D. Schleser (2012). 'El impacto de las políticas laborales contracíclicas sobre el empleo asalariado registrado'. In MTEySS-ILO (eds.), *Macroeconomía, empleo e ingresos: debates y políticas en Argentina frente a la crisis internacional 2008-2009*. Buenos Aires: MTEySS-ILO.
- Chacaltana, J. (2001). 'Reforma en la contratación y el despido en el Perú de los noventa'. *Revista Economía*, 24(48). Lima: Catholic University of Peru.
- Chacaltana, J. (2008). 'Una evaluación del régimen laboral especial para la microempresa en Perú, al cuarto año de vigencia'. Lima: ILO, unpublished.
- Chen, M. (2012). 'The Informal Economy: Definitions, Theories and Policies'. WIEGO Working Paper 1. Cambridge, MA: Women in Informal Employment Globalizing and Organizing.
- Cornia, A. (2012). 'Inequality Trends and their Determinants: Latin America over 1990-2011'. WIDER Working Paper 2012/09. Helsinki: UNU-WIDER.
- Corseuil, C., and M. Foguel (2012) 'Economic Expansion and Increase in Labour Market Formality: A Poaching Approach'. Revista Brasileira de Economia, 66(2): 207–24.
- Cruces, G., S. Galiani, and S. Kidyba (2010). 'Payroll Taxes, Wages and Employment: Identification through Policy Changes'. CEDLAS Working Paper 93. La Plata: CEDLAS.
- de Andrade, G., M. Bruhn, and D. McKenzie (2013). 'A Helping Hand or the Long Arm of the Law? Experimental Evidence on What Governments Can Do to Formalize Firms'. Policy Research Working Paper 6435. Washington, DC: World Bank.
- Delgado, G., A. Querino, A. Campos, F. Vaz, L. Rangel, and M. Stivali (2007). 'Avaliação do Simples: Implicações à Formalização Previdenciária'. Texto para discussão 1277. Brasilia: IPEA.
- De Soto, H. (1986). El otro sendero: la revolución informal. Lima: Editorial Ausonia.
- De Vreyer, P., and F. Roubaud (eds) (2013). Urban Labor Markets in Sub-Saharan Africa. Washington, DC: World Bank. https://doi.org/10.1596/978-0-8213-9781-7
- Dirección General de Estadística, Encuestas y Censos (2017). *Encuesta Continua de Empleo*. Available at: https://www.dgeec.gov.py/datos/encuestas/ece/Poblacion/ECE-2017/Trimestre3/dicc/?dicc=MTQ3 (accessed 11 January 2021).
- ECLAC (2006). La protección social de cara al futuro: acceso, financiamiento y solidaridad (LC/G.2294(SES.31/3)). Santiago de Chile: ECLAC.
- ECLAC (2008). Social Panorama of Latin America 2008 (LC/G.2402). Santiago de Chile: Economic Commission for Latin America and the Caribbean.
- ECLAC (2013). Social Panorama of Latin America 2013 (LC/G.2580). Santiago de Chile: Economic Commission for Latin America and the Caribbean.

- ECLAC (2017). Social Panorama of Latin America 2016. Santiago de Chile: Economic Commission for Latin America and the Caribbean.
- ECLAC and ILO (2014). 'Formalización del empleo y distribución de los ingresos laborales', Coyuntura laboral en América Latina y el Caribe 11, Santiago: Economic Commission for Latin America and the Caribbean and International Labour Organization.
- ECLAC and ILO (2019). 'Coyuntura Laboral en América Latina y el Caribe. Evolución y perspectivas de la participación laboral femenina en América Latina'. Santiago: Economic Commission for Latin America and the Caribbean and International Labour Organization.
- Fajnzylber, P., W. Maloney, and G. Rojas (2009). 'Does Formality Improve Micro-Firm Performance? Quasi-experimental Evidence from the Brazilian SIMPLES Program'. IZA Discussion Paper 4531. Bonn: IZA.
- Fields, G. (1990). 'Labour Market Modelling and the Urban Informal Sector: Theory and Evidence'. In D. Turnham, B. Salomé, and A. Schwarz (eds), *The Informal Sector Revisited*. Paris: Organisation for Economic Co-operation and Development.
- Grimm, M., P. Knorringa, and J. Lay (2012). 'Constrained Gazelles: High Potentials in West Africa's Informal Economy'. World Development, 40(7): 1352–68. https://doi.org/10.1016/j.worlddev.2012.03.009
- Hussmanns, R. (2004). 'Measuring the Informal Economy: from Employment in the Informal Sector to Informal Employment'. Policy Integration Department Working Paper 53. Geneva: ILO.
- IBGE (2016). Pesquisa Mensal de Emprego. Rio de Janeiro: Instituto Brasileiro de Geografia e Estatistica.
- IBGE (2020). Pesquisa Nacional por Amostra de Domicilios Contínua. Available at: https://biblioteca.ibge.gov.br/visualizacao/livros/liv101733\_notas\_tecnicas.pdf (accessed 11 January 2021).
- ILO (1972). 'Employment, Income and Equality: a Strategy for Increasing Productive Employment in Kenya'. Report of an inter-agency team financed by the United Nations Development Programme and organised by the International Labour Office. Geneva: ILO.
- ILO (1993). 'Resolution Concerning Statistics of Employment in the Informal Sector'. The Fifteenth International Conference of Labour Statisticians, Geneva: ILO.
- ILO (2002). Resolution Concerning Decent Work and the Informal Economy. Governing Body, 285th Session. Geneva: International Labour Organization.
- ILO (2014). 'Thematic Labour Overview 1: Transition to Formality in Latin America and the Caribbean' (Revised version) Lima: International Labour Organization, Regional Office for Latin America and the Caribbean.
- ILO (2018). Políticas de Formalización en América Latina. Avances y desafíos'. J. Salazar-Xirinachs and J. Chacaltana (eds.). Lima: Oficina Regional para América Latina y el Caribe.
- INDEC (2019). Encuesta Permanente de Hogares: Diseno de Registro y Estructura para las bases prelinares Hogar y Personas. Buenos Aires: Instituto Nacional de Estadística y Censos.
- INEC (2018). Encuesta Nacional de Empleo, Desempleo y Subempleo. Quito: Instituto Nacional de Estadística y Censos.
- INEGI (2007). Cómo se hace la ENOE. Aguascalientes City: Instituto Nacional de Estadística y Geografía.
- INEI (2020). Encuesta Nacional de Hogares. Instituto Nacional de Estadistica e Informática. Available at: http://iinei.inei.gob.pe/iinei/srienaho/Descarga/DocumentosMetodologicos/2019-151/13\_Diccionario\_PANEL\_2015\_2019.pdf (accessed 11 January 2021).
- Infante, R. (ed) (2011). El desarrollo inclusivo en América Latina y el Caribe. Ensayos sobre políticas de convergencia productiva para la igualdad. Santiago: CEPAL.
- Kanbur, R. (2017). Informality: Causes, Consequences and Policy Responses'. Review of Development Economics, 21(4): 939–61. https://doi.org/10.1111/rode.12321

- Keifman, S., and R. Maurizio (2014). 'Changes in Labour Market Conditions and Policies, and their Impact on Wage Inequality During the Last Decade'. In G. Cornia (ed.), *Falling Inequality in Latin America: Policy Changes and Lessons*. Oxford: UNU-WIDER and Oxford University Press. https://doi.org/10.1093/acprof:oso/9780198701804.003.0012
- Kucera, D., and L. Roncolato, (2008). 'Informal Employment: Two Contested Policy Issues'. *International Labour Review*, 147(4): 321–48. https://doi.org/10.1111/j.1564-913X.2008.00039.x
- Lustig, N., L. Lopez-Calva, and E. Ortiz-Juarez (2013). 'Declining Inequality in Latin America in the 2000s: The Cases of Argentina, Brazil, and Mexico'. *World Development*, 44: 129–41. https://doi.org/10.1016/j.worlddev.2012.09.013
- Maurizio, R. (2015). 'Transitions to Formality and Declining Inequality: Argentina and Brazil in the 2000s'. Journal of Development and Change, 46(5): 1047–79. https://doi.org/10.1111/dech.12195
- Maurizio, R. (2016). 'Labor Market and Income Distribution in Latin America in Times of Economic Growth: Advances and Shortcomings'. In M. Damill, M. Rapetti, and G. Rozewurcell (eds), *Macroeconomics and Development: Essays in Honour of Roberto Frenkel*. New York, NY: Columbia University Press. https://doi.org/10.7312/columbia/9780231175081.003.0008
- Maurizio, R. (2018). 'In-work Poverty in Latin America. Prevalence, Driving Forces and Trends'. In H. Lohmann and I. Marx (eds), *Handbook of Research on In-Work Poverty*. Cheltenham: Edward Elgar Publishing..
- Maurizio, R. (2020). 'Rotación ocupacional y calidad del empleo. El caso de los trabajadores independientes en América Latina'. *Revista Desarrollo Económico*, 60(230) : 27–58.
- Maurizio, R., and G. Vázquez (2015). 'Desigualdad e informalidad en América Latina: el caso del Ecuador'. In V. Amarante, and R. Arim (eds), *Desigualdad e Informalidad*. Un análisis de cinco experiencias latinoamericanas. CEPAL-Ministerio de Asuntos Exteriores de Noruega. https://doi.org/10.18356/a265b870-es
- Maurizio, R., and G. Vázquez (2016). 'Distribution Effects of the Minimum Wage in Four Latin American Countries: Argentina, Brazil, Chile and Uruguay'. *International Labour Review*, 155(1): 97– 131. https://doi.org/10.1111/ilr.12007
- Maurizio, R., and G. Vázquez (2019). 'Formal Salaried Employment Generation and Transition to Formality in Developing Countries: The Case of Latin America'. Employment Working Paper 251. Geneva: ILO.
- Monteiro, J., and J. Assunção (2012). 'Coming Out of the Shadows? Estimating the Impact of Bureaucracy Simplification and Tax Cut on Formality in Brazilian Microenterprises'. *Journal of Development Economics*, 99(1): 105–15. https://doi.org/10.1016/j.jdeveco.2011.10.002
- Perry, G., W. Maloney, O. Arias, P. Fajnzylber, A. Mason, and J. Saavedra-Chanduvi (2007). Informality: Exit and Exclusion. Washington, DC: World Bank. https://doi.org/10.1596/978-0-8213-7092-6
- Ronconi, L. (2010). 'Enforcement and Compliance with Labor Regulations'. *Industrial and Labor Relations* Review, 63(4): 719–36. https://doi.org/10.1177/001979391006300409
- Tannuri-Pianto, M., and D. Pianto (2002). 'Formal-informal Differentials in Brazil: A Semi-parametric Approach'. Paper presented at Graduate School of Economics, Getulio Vargas Foundation.
- Tornarolli, L., D. Battistón, L. Gasparini, and P. Gluzmann (2014). 'Exploring Trends in Labor Informality in Latin America, 1990-2010'. CEDLAS Working Paper 159. La Plata: CEDLAS.
- World Bank (2012). 'Ecuador. Las caras de la informalidad'. Report 67808-EC, Oficina Regional para América Latina y el Caribe. Washington, DC: World Bank.

# Appendix

Table A1: Formality rate from cross-section and panel data

ARGENTINA			BR	AZIL			ECUA	DOR			MEX	KICO			PARA	GUAY			PE	RU				
Groups	CR	OSS	PA	NEL	CRO	OSS	PA	NEL	CRO	OSS	PA	NEL	CRO	DSS	PA	NEL	CRO	DSS	PA	NEL	CRO	DSS	PANE	L
	2003	2019	2003	2019	2003	2015	2003	2015	2004	2019	2004	2019	2005	2019	2005	2019	2010	2017	2010	2017	2004	2018	2004	2018
Education																								
Less comp. sec.	17	13	17	13	29	20	25	21	10	12	9	14	13	7	12	7	8	8	9	9	4	5	4	6
Comp. secondary Incomp. tertiary	20	29	21	31	30	42	32	40	20	31	28	28	21	25	21	25	26	43	24	44	13	26	12	28
Complete tertiary	15	23	14	22	12	21	16	22	11	19	14	17	18	19	20	18	4	3	2	3	21	30	18	28
TOTAL	52	65	52	66	70	83	73	82	41	62	52	59	52	52	53	51	38	55	35	56	37	61	34	62
Gender																								
Women	20	28	20	29	30	38	32	39	17	26	21	24	20	21	20	19	17	23	13	25	13	24	11	24
Men	32	37	32	37	40	45	41	44	24	36	31	35	32	31	33	32	21	31	22	31	25	37	23	38
TOTAL	52	65	52	66	70	83	73	82	41	62	52	59	52	52	53	51	38	55	35	56	37	61	34	62
Age																								
Less than 25	5	5	5	3	13	11	14	11	4	6	5	5	9	7	9	6	5	8	5	9	2	7	1	8
25–45	30	37	31	41	42	48	38	41	23	36	30	35	32	30	32	29	24	35	24	34	23	34	22	32
More than 45	17	23	17	21	15	24	21	30	13	20	17	20	11	16	13	16	8	12	6	13	12	21	11	22
TOTAL	52	65	52	66	70	83	73	82	41	62	52	59	52	52	53	51	38	55	35	56	37	61	34	62
Firm size																								
Less 6 employees	6	9	7	8	7	6	6	6	3	7	2	7	3	2	3	2	2	3	2	3	3	7	3	7
6–40 employees	18	21	19	22	4	3	3	4	9	13	13	12	19	18	19	18	12	17	13	20	7	13	6	14
More than 40	28	35	27	35	59	74	63	72	29	43	37	41	30	31	31	30	21	35	18	33	28	41	25	40
TOTAL	52	65	52	66	70	83	73	82	41	62	52	59	52	51	54	50	35	55	33	56	37	61	34	61
Tenure																								
1 year or less	7	5	7	5	14	15	14	10	18	29	21	26	8	8	8	7	7	9	7	10	8	21	8	21
>1–5 years	15	19	15	19	31	40	31	42	7	15	10	14	19	20	18	20	12	18	12	20	13	19	10	19
More than 5	30	41	30	41	25	29	28	30	15	19	22	19	25	24	27	25	19	27	16	26	20	21	21	22
TOTAL	52	65	52	66	70	83	73	82	41	62	52	59	52	52	54	51	38	55	35	56	42	61	39	62

Table A2: Distribution of individuals across employment and work statuses

	Working-age population	Labour force	All workers	Informal workers	Self- employment	Wage employment
			ARG	ENTINA		
Formal wage employees	28	42	46	-	-	62
Upper-tier informal wage emp.	7	10	11	23	-	15
Lower-tier informal wage emp.	10	15	17	34	-	23
Formal self-employed	3	4	5	-	18	-
Informal self-employed	12	19	21	42	82	-
Unemployed	7	10	-	-	-	-
Inactive	34	-	-	-	-	-
TOTAL	100	100	100	100	100	100
			BF	RAZIL		
Formal wage employees	31	48	53	-	-	76
Upper-tier informal wage emp.	5	7	8	20	-	12
Lower-tier informal wage emp.	5	8	9	21	-	12
Formal self-employed	3	5	5	-	17	-
Informal self-employed	14	22	25	59	83	-
Unemployed	6	10	-	-	-	-
Inactive	35	-	-	-	-	-
TOTAL	100	100	100	100	100	100
			ECI	JADOR		
Formal wage employees	25	39	41	-	-	63
Upper-tier informal wage emp.	6	9	10	17	-	15
Lower-tier informal wage emp.	9	14	15	27	-	23
Formal self-employed	2	3	3	-	10	-
Informal self-employed	19	29	31	56	90	-
Unemployed	4	6	-	-	-	-
Inactive	34	-	-	-	-	-
TOTAL	100	100	100	100	100	100

			М	EXICO		
Formal wage employees	24	38	40	-	-	57
Upper-tier informal wage emp.	7	11	11	20	-	16
Lower-tier informal wage emp.	12	18	19	34	-	27
Formal self-employed	2	4	4	-	14	-
Informal self-employed	16	24	26	46	86	-
Unemployed	3	5	-	-	-	-
Inactive	36	-	-	-	-	-
TOTAL	100	100	100	100	100	100
			PAF	RAGUAY		
Formal wage employees	23	31	35	-	-	48
Upper-tier Informal wage emp.	12	16	18	29	-	25
Lower-tier Informal wage emp.	13	18	20	32	-	27
Formal self-employed	1	2	2	-	8	-
Informal self-employed	16	22	24	39	92	-
Unemployed	8	11	-	-	-	-
Inactive	25	-	-	-	-	-
TOTAL	100	100	100	100	100	100
			F	PERU		
Formal wage employees	17	23	25	-	-	47
Upper-tier informal wage emp.	10	13	14	20	-	26
Lower-tier informal wage emp.	10	14	15	22	-	28
Formal self-employed	4	5	5	-	12	-
Informal self-employed	27	36	40	58	88	-
Unemployed	7	9	-	-	-	-
Inactive	25	-	-	-	-	-
TOTAL	100	100	100	100	100	100

#### Table A3: Descriptive statistics by work status

		Argentina			Brazil			Ecuador	
Education	SI	SC-TI	TC	SI	SC-TI	TC	SI	SC-TI	TC
Formal wage employees	28	43	29	28	47	25	22	50	28
Upper-tier informal wage emp.	50	35	15	33	48	20	52	39	9
Lower-tier informal wage emp.	64	32	4	68	28	3	65	33	3
Formal self-employed	4	11	85	5	13	82	3	7	90
Informal self-employed	55	40	4	55	40	5	58	40	3
TOTAL	41	38	21	36	43	21	42	42	17
Gender	Women	Men		Women	Men		Women	Men	
Formal wage employees	41	59		46	54		41	59	
Upper-tier informal wage emp.	42	58		45	55		27	73	
Lower-tier informal wage emp.	56	44		69	31		37	63	
Formal self-employed	45	55		42	58		36	64	
Informal self-employed	34	66		37	63		45	55	
TOTAL	43	57		46	54		40	60	
Age	<26	25–46	>46	<26	25–46	>46	<26	25–46	>46
Formal wage employees	9	58	33	17	49	33	10	59	31
Upper-tier informal wage emp.	24	53	22	33	39	28	24	55	20
Lower-tier informal wage emp.	25	47	28	18	38	44	23	52	25
Formal self-employed	2	49	49	2	41	57	2	54	44
Informal self-employed	8	44	47	5	36	59	8	47	46
TOTAL	13	52	35	16	45	40	12	54	34

		Mexico			Paraguay			Peru	
Education	SI	SC-TI	TC	SI	SC-TI	TC	SI	SC-TI	TC
Formal wage employees	19	45	36	20	74	6	12	40	48
Upper-tier informal wage emp.	33	46	21	42	55	3	28	52	20
Lower-tier Informal wage emp.	49	46	6	64	35	0	40	49	10
Formal self-employed	4	7	89	13	31	55	2	5	92
Informal self-employed	47	46	6	54	46	1	47	49	4
TOTAL	33	44	23	41	55	4	32	45	23
Gender	Women	Men		Women	Men		Women	Men	
Formal wage employees	39	61		40	60		35	65	
Upper-tier informal wage emp.	36	64		31	69		40	60	
Lower-tier informal wage emp.	42	58		52	48		46	54	
Formal self-employed	39	61		44	56		41	59	
Informal self-employed	43	57		46	54		52	48	
TOTAL	40	60		42	58		44	56	
Age	<26	25–46	>46	<26	25–46	>46	<26	25–46	>46
Formal wage employees	14	58	27	14	66	20	10	59	31
Upper-tier informal wage emp.	29	51	21	30	58	12	34	50	16
Lower-tier informal wage emp.	28	47	25	28	53	19	35	47	19
Formal self-employed	3	53	44	1	57	43	5	59	36
Informal self-employed	12	45	43	7	52	40	16	45	39
TOTAL	18	52	31	18	58	24	19	50	31

	Arger	ntina	Bra	azil	Ecu	ador
Work status	Weekly hours	Hourly wages	Weekly hours	Hourly wages	Weekly hours	Hourly wages
Formal wage employees	41.0	6.2	41.1	6.7	43.5	3.4
Upper-tier informal wage emp.	35.4	3.1	36.9	5.4	43.0	2.0
Lower-tier informal wage emp.	32.7	2.7	36.5	2.7	40.0	1.6
Formal self-employed	41.9	7.2	41.2	17.1	39.9	4.9
Informal self-employed	42.6	3.6	41.1	5.0	38.6	2.3
	Mex	ico	Para	guay	Pe	eru
Work status	Weekly hours	Hourly wages	Weekly hours	Hourly wages	Weekly hours	Hourly wages
Formal wage employees	45.7	23.5	47.1	7249.6	44.9	6.8
Upper-tier informal wage emp.	43.7	19.0	47.4	6612.2	44.3	4.1
Lower-tier informal wage emp.	41.9	14.1	43.8	4139.4	43.0	3.2
Formal self-employed	42.4	31.8	47.5	13833.6	41.5	10.1
Informal self-employed	40.6	16.2	44.0	6889.3	43.2	4.1

Table A5: Year-to-year conditional probabilities of transitions (%)

					Emple	oyment stat	us in Year t	+ 1		
Country	Employment sta	atus in Year t	Formal wage	Informa emplo	al wage byees	Self-er	nployed	Un- employed	Inactive	Total
			employees	Upper-tier	Lower-tier	Formal	Informal			
Argentina	Formal wage er	mployees	87.4	2.7	2.4	0.8	1.8	1.9	2.9	100
	Informal wage	Upper-tier	18.7	39.4	14.3	2.2	8.3	5.9	11.1	100
	employees	Lower-tier	9.4	9.0	46.3	0.9	11.1	7.3	16.0	100
	Self-employed	Formal	8.2	5.2	3.0	59.0	17.0	2.1	5.6	100
		Informal	4.1	4.2	8.5	3.8	62.9	4.5	11.9	100
	Unemployed		10.4	8.5	13.8	1.3	11.2	26.5	28.4	100
	Inactive		2.5	2.8	5.6	0.5	4.3	6.1	78.1	100
Brazil	Formal wage er	mployees	79.0	2.6	2.2	0.9	2.6	5.8	6.9	100
	Informal wage	Upper-tier	21.4	33.3	7.2	3.0	9.3	11.1	14.5	100
	employees	Lower-tier	7.0	2.9	46.4	0.6	12.4	9.6	21.1	100
	Self-employed	Formal	3.7	2.4	1.0	71.2	13.8	2.6	5.3	100
		Informal	1.8	1.2	3.5	2.6	69.0	5.2	16.6	100
	Unemployed		7.1	2.9	5.5	1.4	10.4	41.8	30.9	100
	Inactive		1.2	0.8	2.2	0.4	5.6	8.8	81.1	100
Ecuador	Formal wage er	mployees	85.1	2.8	2.1	1.0	3.3	2.1	3.5	100
	Informal wage	Upper-tier	22.1	33.9	18.5	1.5	12.1	4.4	7.6	100
	employees	Lower-tier	9.4	10.8	44.5	0.7	19.8	4.5	10.3	100
	Self-employed	Formal	12.8	4.6	3.3	47.0	22.3	2.5	7.6	100
		Informal	3.8	3.4	8.2	2.3	67.0	2.2	13.1	100
	Unemployed		15.1	7.0	13.6	2.9	13.5	18.2	29.7	100
	Inactive		2.9	1.9	4.0	0.6	8.4	5.0	77.2	100
Mexico	Formal wage er	mployees	77.8	6.1	3.8	1.1	2.7	2.7	5.8	100
	Informal wage	Upper-tier	26.5	30.9	16.3	1.9	7.9	4.1	12.4	100
	employees	Lower-tier	8.5	9.2	49.1	0.8	13.4	3.3	15.7	100
	Self-employed	Formal	8.9	4.8	4.7	50.8	14.6	2.3	13.9	100
		Informal	4.1	3.5	10.5	2.4	59.6	1.6	18.2	100
	Unemployed		21.9	9.9	14.4	2.4	9.0	14.9	27.6	100
	Inactive		3.8	2.7	5.7	0.8	7.8	2.5	76.6	100
Paraguav	Formal wage er	mplovees	86.8	5.0	1.5	0.2	2.5	2.1	1.9	100
	Informal wage	Upper-tier	15.6	50.9	13.0	0.6	8.6	7.1	4.2	100
	employees	Lower-tier	5.3	12.5	52.9	0.1	10.7	7.5	11.0	100
	Self-employed	Formal	5.2	5.0	2.2	36.8	42.7	6.6	1.5	100
		Informal	1.9	3.1	7.3	2.7	67.8	5.4	11.8	100
	Unemployed	- Internation	9.8	16.5	18.0	0.5	7.8	27.8	19.6	100
	Inactive		2.3	4.0	8.0	0.2	7.9	9.5	68.2	100
Peru	Formal wage er	mplovees	78.9	6.3	2.3	2.0	4.0	3.5	3.0	100
	Informal wage	Upper-tier	20.5	36.9	13.1	2.0	11.3	5.7	10.6	100
	employees	Lower-tier	62	14 3	43.5	1.0	17.4	5.4	12 3	100
	Self-employed	Formal	10.2	3.6	4.8	48.4	20.3	3.4	8 1	100
		Informal	29	4 N	0 5.6	3.0	70.5	33	10.6	100
	Unemployed	monna	11 3	8.8	11 4	2.5	12.2	20.8	33.0	100
	Inactive		28	5.0	61	1 3	12.2	Q 2	62.1	100
	macuve		2.0	5.1	0.1	1.0	10.0	5.2	02.1	100

Note: each row indicates work status in the initial year and each column indicates work status in the next year. The likelihood of staying in the same employment status conditional on the base year employment status is highlighted in grey.

Table A6: Year-to-year transitions up and down the job ladder (%)

Argentina	From Wor	k Status 3	From Wo	rk Status 2	From Work Status 1		
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2	
Age							
Age	-0.0345**	-0.0313***	-0.0105	0.0616***	-0.129***	-0.201***	
Age*Age	-0.000105	0.000148	0.000150	-0.000899***	0.00155***	0.00206***	
Gender (female)							
Male	0.161**	0.148***	0.0358	0.127*	-0.0265	-0.0312	
Education (less comp. secondary)							
Comp. secondary. Incomp. tertiary	-0.112**	0.582***	-0.0887	0.635***	-0.211***	-0.745***	
Complete tertiary	0.126	2.090***	-1.309***	1.149***	0.111*	-0.315***	
Sector of activity (manufacture)							
Construction	-0.185**	-0.0701	0.250**	-0.232	1.193***	1.041***	
Trade	-0.383***	0.0165	0.0371	-0.0793	0.376***	0.280**	
Transport	0.325***	0.483***	-0.285**	-0.510***	-0.321***	0.212	
Financial services	-0.114	0.859***	-0.109	-0.314**	-0.268***	0.322***	
Personal services	0.651***	0.955***	-0.665***	-0.335**	-0.607***	0.414***	
Domestic services	-0.973***	0.639***	0.962	-11.05	1.296***	-1.185***	
Public sector	-0.671	2.202***	-1.585***	-0.339***	-1.825***	-0.134	
Other activity	-0.594***	0.171*	-0.286**	-0.262*	0.0258	0.359***	
Region FE	YES	YES	YES	YES	YES	YES	
Constant	-0.666**	-1.688***	0.121	-1.825***	-0.261	1.219***	
Observations	23,890	23,890	5,926	5,926	39,805	39,805	
Brazil	From Wor	k Status 3	From Wo	rk Status 2	From Wor	rk Status 1	
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2	
Age							
Age	-0.111***	-0.0476***	0.113***	0.0223***	0.0217***	-0.143***	
Age*Age	0.000930***	0.000143***	-0.00112***	-0.000403***	-1.24e-05	0.00162***	
Gender (female)							
Male	0.183***	0.236***	0.0765***	0.0506***	0.239***	-0.0128	
Education (less comp. secondary)							
Comp. secondary. Incomp. tertiary	0.0812***	0.398***	-0.237***	0.207***	-0.160***	-0.0206	
Complete tertiary	0.181***	1.694***	-1.485***	0.549***	-0.209***	0.149***	
Sector of activity (manufacture)							

Age*Age	0.000930***	0.000143***	-0.00112***	-0.000403***	-1.24e-05	0.00162***
Gender (female)						
Male	0.183***	0.236***	0.0765***	0.0506***	0.239***	-0.0128
Education (less comp. secondary)						
Comp. secondary. Incomp. tertiary	0.0812***	0.398***	-0.237***	0.207***	-0.160***	-0.0206
Complete tertiary	0.181***	1.694***	-1.485***	0.549***	-0.209***	0.149***
Sector of activity (manufacture)						
Construction	-0.440***	0.101***	0.572***	0.0218	0.888***	0.175***
Trade	-0.160***	-0.0179	0.254***	0.0900***	0.526***	0.0725**
Transport	0.314***	0.0234	0.0869	0.0213	-0.0127	-0.102**
Financial services	0.620***	0.556***	-0.316***	-0.137***	0.116***	0.161***
Personal services	1.375***	1.043***	-0.451***	-0.0808**	-0.376***	0.0590
Domestic services	-0.922***	0.746***			1.380***	-1.305***
Public sector			0.0183	-0.392***	0.402***	0.486***
Other activity	-0.0585	-0.185***	-0.488***	-0.176***	-0.104**	0.0188
Region FE	YES	YES	YES	YES	YES	YES
Constant	0.191	-0.380***	-2.600***	-0.173*	-3.745***	-0.430***
Observations	178,814	178,814	58,208	58,208	439,990	439,990

Ecuador	From Wor	k Status 3	atus 3 From Work Status 2			From Work Status 1		
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2		
Age						-		
Age	-0.0773***	-0.0379***	0.0290	0.0290	-0.0420***	-0.124***		
Age*Age	0.000462***	0.000144	-0.000274	-0.000274	0.000692***	0.00123***		
Gender (female)								
Male	0.816***	0.266***	0.151	0.151	0.212***	0.106		
Education (less comp. secondary)								
Comp. secondary. Incomp. tertiary	-0.215***	0.997***	-0.0700	-0.0700	-0.255***	-0.602***		
Complete tertiary	0.00954	2.886***	-0.500***	-0.500***	0.0553	-0.527***		
Sector of activity (manufacture)								
Construction	0.463***	0.0680	0.485***	0.485***	1.050***	0.595***		
Trade	-0.346***	-0.231***	0.405***	0.405***	0.475***	-0.225*		
Transport	-0.366***	-0.151	0.540***	0.540***	0.780***	-0.193		
Financial services	0.489***	0.818***	-0.0185	-0.0185	0.171*	-0.339**		
Personal services	0.997***	1.103***	-0.578***	-0.578***	-0.354***	0.120		
Domestic services	-0.101	0.798***			0.595***	-1.982***		
Public sector	///	///	///	///	///	///		
Other activity	0.346***	-0.299***	-0.662***	-0.662***	0.0743	0.272**		
Region FE	YES	YES	YES	YES	YES	YES		
Constant	-0.781***	-1.833***	-0.772**	-0.772**	-2.038***	0.0560		
Observations	25,202	25,202	4,934	4,934	18,510	18,510		
Mexico	From Wor	k Status 3	From Wo	rk Status 2	From Wor	k Status 1		
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2		
Age								
Age	-0.0407***	-0.0176***	0.00852**	0.0211***	-0.0520***	-0.0947***		
Age*Age	7.96e-05*	-4.20e-05	-4.13e-05	-0.000226***	0.000725***	0.000974***		
Gender (female)								
Male	0.456***	0.364***	0.172***	0.0325*	0.261***	0.104***		
Education (less comp. secondary)								
Comp. secondary. Incomp. tertiary	0.0149	0.508***	-0.00606	0.455***	-0.217***	-0.381***		
Complete tertiary	0.577***	2.342***	-0.887***	1.003***	-0.631***	-0.622***		
Sector of activity (manufacture)								
Construction	-0.0964***	0.178***	0.892***	0.0353	1.252***	0.993***		
Trade	-0.494***	-0.118***	0.312***	0.0893***	0.590***	0.271***		
Transport	-0.476***	-0.143***	0.0741*	-0.0509	0.227***	0.280***		
Financial services	0.220***	0.500***	0.0182	0.185***	0.495***	0.519***		
Personal services	0.527***	0.576***	-0.544***	-0.351***	0.475***	0.803***		
Domestic services	-0.745***	-0.0307			2.712***	0.0268		
Public sector	1.145***	1.154***	-0.921***	-0.121***	-0.810***	0.127***		
Other activity	-0.207***	-0.247***	0.235***	-0.486***	0.748***	0.564***		
Region FE	YES	YES	YES	YES	YES	YES		
<b>a</b>	4 000***	1 050***	-0 370***	-0 640***	1 /00***	0 690***		

344,339

344,339

Observations

85,893

85,893

412,095

412,095

Paraguay	From Wor	ork Status 3 From Wo		K Status 2     From Work Status		k Status 1
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2
Age						
25–45	-0.970***	-0.254	-0.218	0.00287	0.335	-0.890**
More than 45	-1.327***	-0.429	0.0585	-0.273	0.0977	-1.249**
Gender (female)						
Male	0.370	0.515*	0.169	0.283	0.0498	-0.317
Education (less comp. secondary)						
Comp. secondary. Incomp. tertiary	-0.0288	1.537***	-0.0171	0.643**	-0.747**	0.192
Complete tertiary	-11.46	2.797***	-13.92	1.042	0.683*	0.474
Sector of activity (manufacture)						
Construction	0.636	-0.669	0.607	-0.441	1.029*	-0.384
Trade	0.0178	-0.175	-0.286	-0.656	0.519	-0.673
Transport	0.298	-0.748	0.110	-0.648	0.590	0.382
Financial services	0.737	-0.702	-0.864*	-0.459	-0.276	-0.797
Personal services	0.771	-0.395	-0.728	-1.320**	-0.613	-1.141**
Domestic services	0.0521	-0.637			18.50	-0.0955
Public sector	0.950	-1.843**	-0.126	-2.783***	0.392	-0.580
Other activity						
Region FE	NO	NO	NO	NO	NO	NO
Constant	-2.076***	-3.112***	-0.662	-0.996*	-2.649***	-1.476**
Observations	1,048	1,048	443	443	1,002	1,002

Peru	From Work Status 3		From Wor	From Work Status 2		From Work Status 1	
	To WS 2	To WS 1	To WS 3	To WS 1	To WS 3	To WS 2	
Age							
Age	-0.0437***	0.0125	0.0424*	0.0647**	-0.0171	-0.110***	
Age*Age	-7.64e-05	-9.34e-05	-0.000459	-0.000678*	0.000340	0.00101***	
Gender (female)							
Male	0.286***	0.580***	-0.130	0.417***	0.0308	0.135	
Education (less comp. secondary)							
Comp. secondary. Incomp. tertiary	0.147**	1.021***	-0.234**	0.941***	-0.546***	-0.221*	
Complete tertiary	0.880***	2.644***	-0.714***	1.838***	-0.477***	-0.756***	
Sector of activity (manufacture)							
Construction	0.525***	0.621***	0.437***	0.683***	0.638***	0.402**	
Trade	-0.591***	-0.224**	0.135	0.575***	0.485***	0.212	
Transport	-0.495***	-0.00623	0.471**	-0.157	0.797***	-0.162	
Financial services	0.264*	0.419***	-0.0387	0.419**	-0.162	-0.192	
Personal services	1.226***	0.387	-0.999***	0.176	-0.346**	-0.126	
Domestic services	-0.380**	0.118			1.395***	-0.838	
Public sector	-20.29	4.919*	-0.337**	0.324**	-2.269***	-0.296**	
Other activity	0.192	0.550***	-0.105	-0.0635	0.164	0.745***	
Region FE	YES	YES	YES	YES	YES	YES	
Constant	-0.828***	-4.074***	-0.828*	-3.487***	-1.270***	0.386	
Observations	17,271	17,271	3,426	3,426	11,825	11,825	

Note: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

# Table A7: Wage change regressions

	Origin work status				
Argentina	Lower-tier / informal self- employed	Upper-tier informal	Formal wage- employed and self-employed		
Log of hourly wage in t	-0.731***	-0.696***	-0.575***		
Age (less than 25)					
25–45	0.104***	0.0417**	0.0718***		
More than 45	0.153***	0.105***	0.109***		
Gender (female)					
Male	0.129***	0.0663***	0.0453***		
Education (less comp. secondary)					
Comp. secondary. Incomp. tertiary	0.178***	0.112***	0.131***		
Complete tertiary	0.326***	0.428***	0.292***		
Sector of activity (manufacture)					
Construction	-0.0206	-0.0408	-0.0227*		
Trade	-0.136***	-0.00606	-0.0585***		
Transport	-0.0104	-0.0626**	0.00977		
Financial services	0.140***	0.122***	0.0411***		
Personal services	0.123***	0.118***	0.0303***		
Domestic services	0.0334*	0.782	-0.122***		
Public sector	-0.177	0.0558**	0.0726***		
Other activity	-0.0117	0.0503*	0.0326***		
Region FE	YES	YES	YES		
Years FE	YES	YES	YES		
Dummy to Status 35		-0.131***	-0.259***		
Dummy to Status 2	0.0683***		-0.133***		
Dummy to Status 14	0.178***	0.163***			
Constant	0.699***	0.801***	0.790***		
Observations	22,642	5,748	40,284		

	Origin work status				
Brazil	Lower-tier / informal self- employed	Upper-tier informal	Formal wage- employed and self- employed		
Log of hourly wage in t	-0.388***	-0.347***	-0.252***		
Age (less than 25)					
25–45	0.0876***	0.0266***	0.0359***		
More than 45	0.100***	0.0647***	0.0757***		
Gender (female)					
Male	0.0899***	0.0657***	0.0560***		
Education (less comp. secondary)					
Comp. secondary. Incomp. terirary	0.128***	0.144***	0.0963***		
Complete tertiary	0.427***	0.411***	0.298***		
Sector of activity (manufacture)					
Construction	-0.00247	0.0177*	0.00593*		
Trade	0.00107	-0.0208***	-0.0306***		
Transport	0.111***	0.00575	-0.00996***		
Financial services	0.131***	0.0450***	0.00579***		
Personal services	0.00441	0.0333***	-0.00968***		
Domestic services	-0.00692		-0.0422***		
Public sector		0.00518	0.0437***		
Other activity	0.0315***	0.0291***	0.00548*		
Region FE	YES	YES	YES		
Years FE	YES	YES	YES		
Dummy to Status 35		-0.0578***	-0.108***		
Dummy to Status 2	0.0109**		-0.0505***		
Dummy to Status 14	0.107***	0.0793***			
Constant	0.0478***	0.161***	0.159***		
Observations	178,814	58,208	439,990		

	Origin work status			
Ecuador	Lower-tier / informal self-employed	Upper-tier informal	Formal wage- employed and self-employed	
Log of hourly wage in t	-0.709***	-0.656***	-0.504***	
Age (less than 25)				
25–45	0.109***	0.0757***	0.0649***	
More than 45	0.110***	0.0855***	0.123***	
Gender (female)				
Male	0.139***	0.0831***	0.0353***	
Education (less comp. secondary)				
Comp. secondary. Incomp. tertiary	0.130***	0.119***	0.153***	
Complete tertiary	0.454***	0.351***	0.357***	
Sector of activity (manufacture)				
Construction	0.105***	0.0723***	0.0456**	
Trade	0.0110	-0.0348	-0.0230**	
Transport	0.0771***	0.0228	0.0136	
Financial services	0.121***	0.150***	0.0104	
Personal services	0.0884**	0.0228	0.0225	
Domestic services	-0.00546		-0.0454**	
Public sector				
Other activity	-0.00410	-0.0146	0.0505***	
Region FE	NO	NO	NO	
Years FE	YES	YES	YES	
Dummy to Status 35		-0.0701***	-0.129***	
Dummy to Status 2	0.0240		-0.114***	
Dummy to Status 14	0.205***	0.132***		
Constant	-0.0481*	-0.0302	0.247***	
Observations	24,043	4,892	18,250	

	Origin work status				
Mexico	Lower-tier / informal self- employed	Upper-tier informal	Formal wage- employed and self-employed		
Log of hourly wage in t	-0.655***	-0.565***	-0.495***		
Age (less than 25)					
25–45	0.123***	0.0670***	0.0505***		
More than 45	0.118***	0.0601***	0.0710***		
Gender (female)					
Male	0.122***	0.0679***	0.0530***		
Education (less comp. secondary)					
Comp. secondary. Incomp. tertiary	0.0735***	0.0697***	0.0676***		
Complete tertiary	0.279***	0.366***	0.286***		
Sector of activity (manufacture)					
Construction	0.0820***	0.0933***	0.0704***		
Trade	-0.0133**	0.00918	-0.0471***		
Transport	-0.0241***	0.0378***	0.0459***		
Financial services	0.123***	0.0563***	0.0387***		
Personal services	0.255***	0.117***	0.0697***		
Domestic services	0.0417***		0.0175		
Public sector	1.279**	0.104***	0.169***		
Other activity	-0.0326***	0.0221***	0.0400***		
Region FE	YES	YES	YES		
Years FE	YES	YES	YES		
Dummy to Status 35		-0.0422***	-0.0906***		
Dummy to Status 2	0.00482		-0.0839***		
Dummy to Status 14	0.0893***	0.0710***			
Constant	1.531***	1.400***	1.303***		
Observations	166,385	48,999	216,002		

	Origin work status				
Paraguay	Lower-tier / informal self- employed	Upper-tier informal	Formal wage- employed and self-employed		
Log of hourly wage in t	-0.549***	-0.465***	-0.382***		
Age (less than 25)					
25–45	0.0910	0.0147	0.0649		
More than 45	0.130**	0.00934	0.147***		
Gender (female)					
Male	0.0594	-0.00394	-0.0141		
Education (less comp. sec)					
Comp. secondary. Incomp. tertiary	0.249***	0.172***	0.146***		
Complete tertiary	0.394*	0.592***	0.300***		
Sector of activity (manufacture)					
Construction	-0.00751	-0.00628	-0.0847		
Trade	-0.0976	-0.0649	-0.0892		
Transport	0.105	0.0207	0.0686		
Financial services	0.0442	0.0576	0.0401		
Personal services	-0.0728	0.0998	0.0567		
Domestic services	-0.266***		-0.346		
Public sector	-0.0933	0.101	-0.0623		
Other activity					
Region FE	NO	NO	NO		
Years FE	YES	YES	YES		
Dummy to Status 35		-0.000879	0.0572		
Dummy to Status 2	0.104		-0.0364		
Dummy to Status 14	0.198**	0.0366			
Constant	4.310***	3.822***	3.123***		
Observations	1,065	471	1,057		

	Origin work status			
Peru	Lower-tier / informal self- employed	Upper-tier informal	Formal wage- employed and self-employed	
Log of hourly wage in t	-0.682***	-0.569***	-0.444***	
Age (less than 25)				
25–45	0.0928***	-0.0495*	0.0142	
More than 45	0.0206	-0.0648*	0.0127	
Gender (female)				
Male	0.355***	0.157***	0.108***	
Education (less comp. secondary)				
Comp. secondary. Incomp. tertiary	0.109***	0.0288	0.0429**	
Complete tertiary	0.349***	0.265***	0.201***	
Sector of activity (manufacture)				
Construction	0.135***	0.242***	0.180***	
Trade	-0.0447*	0.0850**	-0.0223	
Transport	-0.0219	0.159***	0.0411	
Financial services	0.184***	0.211***	0.110***	
Personal services	-0.178**	0.226***	-0.0199	
Domestic services	-0.0490		0.0326	
Public sector	-0.233	0.226***	0.0844***	
Other activity	-0.0893**	0.0337	0.0375	
Region FE	YES	YES	YES	
Years FE	YES	YES	YES	
Dummy to Status 35		0.00841	-0.169***	
Dummy to Status 2	0.0810***		-0.0480**	
Dummy to Status 14	0.277***	0.156***		
Constant	0.103**	0.246***	0.271***	
Observations	15,096	3,402	11,997	

Note: \*\*\* p<0.01, \*\* p<0.05, \* p<0.1